Author's response to reviews

Title: Drug Utilization Evaluation of Cefepime in the First Affiliated Hospital of Bengbu Medical College: A Retrospective and Prospective Analysis

Authors:

Shi Qingping (sir_shi@126.com)
Ding Feng (958387264@qq.com)
Sang Ran (13855288331@163.com)
Liu Yan (1055237127@qq.com)
Yuan Haoyu (302989860@qq.com)

Version: 2 Date: 15 August 2012

Author's response to reviews: see over
Details of the revisions

Please include a statement regarding the ethical approval sought for your work within the manuscript.

Please copyedit your manuscript to improve the standard of language used throughout.

Please expand your methods section as at present it is unclear what you have done. Please expand on both the retrospective and prospective sections and include much greater detail on how these were carried out.

Please include a statement regarding any informed consent that was obtained for this study.